

MDPI

Article

# Do IFRS Disclosure Requirements Reduce the Cost of Equity Capital? Evidence from European Firms

Ghouma Ghouma <sup>1</sup>, Hamdi Becha <sup>1</sup>, Maha Kalai <sup>1</sup>, Kamel Helali <sup>1</sup> and Myriam Ertz <sup>2,\*</sup>

- Research Laboratory in Competitiveness, Commercial Decisions and Internationalization (CODECI), Faculty of Economics and Management of Sfax, University of Sfax, Aerodrome Road Km 4.5, Sfax 3018, Tunisia; ghoumaa.ghouma@gmail.com (G.G.); bechahamdi@yahoo.com (H.B.); maha.kalai@fsegs.usf.tn (M.K.); kamel.helali@fsegs.usf.tn (K.H.)
- <sup>2</sup> Laboratory of Research on New Forms of Consumption (LaboNFC), Department of Economics and Administrative Sciences, University of Quebec at Chicoutimi, 555, Boulevard de l'Université, Saguenay, QC G7H 2B1, Canada
- \* Correspondence: myriam\_ertz@uqac.ca; Tel.: +1-41-8545-5011

**Abstract:** This study analyzes the impact of adopting International Financial Reporting Standards (IFRS) on the cost of equity capital for firms listed on STOXX Europe 600 using a sample of 9773 firm-year observations between 1994 and 2022. We estimate the cost of equity capital using the modified price—earnings—growth ratio model and employ the GMM system to investigate the effect of IFRS Standards on the cost of equity capital. Our results indicate that IFRS adoption reduces firms' cost of equity capital. We performed various sensitivity analyses to ensure the reliability of our results. Overall, this study contributes to the extant literature on the cost of equity capital implications of IFRS adoption and provides valuable insights for investors, regulators, and policymakers.

Keywords: cost of equity capital; IFRS; European firms; STOXX Europe 600; GMM-system

JEL Classification: C23; M40; M41; O52



Citation: Ghouma, Ghouma, Hamdi Becha, Maha Kalai, Kamel Helali, and Myriam Ertz. 2023. Do IFRS Disclosure Requirements Reduce the Cost of Equity Capital? Evidence from European Firms. *Journal of Risk* and Financial Management 16: 374. https://doi.org/10.3390/jrfm16080374

Academic Editor: Thanasis Stengos

Received: 26 June 2023 Revised: 27 July 2023 Accepted: 8 August 2023 Published: 15 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/).

## 1. Introduction

The smooth flow of capital and investments between countries is vital for economies, investors, and lenders. To assess cross-border investments, investors worldwide utilize the International Financial Reporting Standards (IFRS) as a universal language, adopted in over 165 countries (Prather-Kinsey et al. 2022). In particular, since 2005, the European Union (EU) has endorsed a regulation allowing listed companies within its member states, including insurance companies and banks, to prepare their consolidated financial statements following IFRS. This move aligns with the International Accounting Standards Board's (IASB) vision of introducing IFRS, an accounting code aiming to establish a unified financial reporting platform on a global scale (Mohsin et al. 2021).

According to AICPA (2005), this regulation marked a significant milestone in financial reporting within Europe, representing the most substantial changes in the past three decades. These changes directly impacted around 7000 companies and had indirect effects on various types of consolidated subsidiaries. The authority governing IFRS is the IASB, an organization dedicated to promoting public interest by fostering long-term economic growth, confidence, and financial stability in the global economy through reliable financial information. In essence, the IFRS supplants Generally Accepted Accounting Principles (GAAP), and, since 2005, the EU and the European Economic Union (EEA) have imposed obligatory requirements for listed companies in both member states and non-member countries (Mager and Meyer-Fackler 2017; Nguyen 2018).

The movement to mandate the adoption of IFRS is considered the most widespread global financial reform in accounting history (Daske et al. 2008). The principle behind these

standards is to improve financial statements' transparency and reliability worldwide and facilitate cross-border investments. Because of this global dimension, it is more difficult and essential to determine the economic consequences of accounting standards in the context of financial regulatory reforms as an increasing number of countries with different levels of development adopt IFRS (Zeff 2012).

Examining these effects has important economic and social implications for European countries, which may impact the domestic and international users of accounting information. Therefore, regulators are interested in knowing whether IFRS adoption may have contributed to reducing the cost of equity capital and, consequently, report an increase in market efficiency and liquidity (Han et al. 2016). Investors are interested in determining whether information asymmetry problems have reduced since IFRS adoption. This indicates decreased information acquisition and verification efforts, allowing for more efficient investment decisions (Diamond and Verrecchia 1991; Ball 2006) and a potential increase in cross-border investment (De Fond et al. 2011).

While previous studies have documented the positive effects of IFRS implementation (i.e., reduction in firms' cost of equity capital), empirical evidence on the role of specific legal disclosure requirements on these financial benefits is lacking. Hellman et al. (2018) argue that non-compliance is significant in both general and specific IFRS disclosures. Therefore, the findings based on IFRS adoption cannot be used to determine the effect of IFRS requirements on the level of disclosure. This creates a gap in the literature that we attempt to fill by explicitly examining the relationship between firm-level IFRS disclosure and its impact on the cost of equity capital. This study sheds light on whether IFRS disclosure requirements benefit users economically and contribute to the disclosure overload debate.

The contributions of this study are two-fold. First, studying the impact of IFRS adoption on the cost of equity capital can help inform policy decisions on financial reporting and accounting standards. Second, the cost of equity capital is an important indicator for companies because it reflects the return investors require to compensate for the risk associated with investing in a particular company. Therefore, understanding the relationship between IFRS and the cost of equity capital can have important implications for both companies and policymakers.

Hence, this study assesses the effect of IFRS on the cost of equity capital for a sample of 337 European firms listed on STOXX 600 Europe in 17 European countries that implemented these standards between 1994 and 2022. To account for cross-sectional dependence among the firms in our sample, we perform CD tests, as suggested by Pesaran (2021). Furthermore, we use the GMM-system technique to examine the relationship between IFRS adoption and the cost of equity capital. The findings from the analysis suggest that there is an inverse relationship between IFRS disclosure requirements and the cost of equity capital. In simpler terms, companies that adhere to higher levels of IFRS disclosure tend to experience lower costs of equity capital.

The remainder of this paper is organized as follows: Section 2 provides an overview of the relevant theoretical and empirical literature and presents the development of our hypotheses. Section 3 details the sample data and methodology used. Section 4 presents the empirical results. Section 5 presents the main conclusions and some policy implications.

## 2. Literature Review and Hypothesis Development

#### 2.1. Theoretical Framework

From a theoretical perspective, separation of ownership gives rise to the need for better governance. Smith (1776) highlighted the agency problem by stating that managers should consider other people's funds rather than their own. He argued that managers could not look after the funds as partners were in a partnership. According to Berle and Means (1932), small shareholders cannot be a controller in large corporations with dispersed ownership because of high costs and low returns (Ali et al. 2019).

In cases where accounting enforcement mechanisms are lacking, a company's corporate governance system and financial reporting incentives, commonly referred to as "corporate characteristics", may significantly impact the determination of incentives for disclosures. According to agency theory, there is an agency relationship in which one party (i.e., principal) delegates work to another (i.e., agent) performing that work on behalf of the principal. Thus, there is a separation of ownership and control of the entity, and it may be expensive or difficult for the principal to verify what the agent is doing because of information asymmetry (Eisenhardt 1989; Jensen and Meckling 1976).

The application of corporate governance principles is a monitoring cost that can be used to curb the information asymmetry caused by agency relationships. For instance, Fama and Jensen (1983) claimed that the role of the board of directors can be used as an information system to monitor shareholders' opportunism toward top executives. Further, Eisenhardt (1989) posited that when the board provides quality financial information (through, for instance, compliance with IFRS disclosure requirements), top executives are more likely to behave consistently with shareholders' interests.

According to Damak-Ayadi et al. (2020), the adoption of IFRS for SMEs' standards in various countries can be attributed to two main theories: the neo-institutional theory, as proposed by DiMaggio and Powell (1997), and the economic theory of networks, as proposed by Katz and Shapiro (1985). According to DiMaggio and Powell (1997), companies that internationalize their operations tend to gain increased legitimacy in the eyes of their stakeholders and the broader business community. Déjean and Saboly (2006) further argued that this quest for organizational legitimacy plays a significant role in influencing firms to adopt specific practices or standards, such as IFRS for SMEs. As a result, firms may embrace these standards not only for their inherent benefits but also to align themselves with prevailing norms and gain acceptance in their international business engagements. As highlighted by Meyer and Rowan (1977), organizations facing environmental constraints should actively employ mechanisms of legitimacy. By doing so, these organizations can establish a favorable image and gain acceptance within their societal and business environments. Adopting mechanisms of legitimacy can involve embracing widely recognized standards, like IFRS, to showcase their commitment to transparency, accountability, and responsible financial reporting.

DiMaggio and Powell (1983) argued that legitimacy is achieved through the concept of "institutional isomorphism". They proposed that a country's full adoption of IFRS can be explained by three types of isomorphism. The first is coercive isomorphism, which refers to the institutional pressures on economic actors to adopt IFRS. Mantzari et al. (2017) defined coercive pressures as occurring when external powerful parties, such as the state and other constituents upon which an organization is dependent, force the adoption of an organizational practice or element, usually by using sanctions. On the other hand, Reichborn-Kjennerud et al. (2019) defined coercive pressure as the social pressure to follow existing societal norms. They highlighted that norms may be formal or informal. Formal coercive norms are based on laws and regulations, while informal coercive pressure includes media and public expectations. The impetus behind the adoption of IFRS can be attributed to the regulatory system and influential international financing organizations, such as the World Bank and International Monetary Fund (IMF), as pointed out by Judge et al. (2010).

Another factor influencing the adoption of IFRS is mimetic isomorphism, where organizations imitate the practices of more efficient counterparts when they face uncertainty in their environment and have ambiguous objectives. Meyer and Rowan (1977) proposed that organizations facing uncertain environments can effectively and economically navigate these challenges by adopting a strategy of imitating the behaviors of successful organizations. In simpler terms, when organizations encounter uncertainties or complexities in their operating environment, they can increase their chances of success by emulating the practices and strategies of established and prosperous companies. By imitating successful

organizations, they can draw upon proven methods and approaches, reducing the risks associated with experimentation and trial-and-error.

Meyer and Rowan (1977) suggested that this imitative approach allows organizations to benefit from the experiences and lessons learned by others, enabling them to adapt more efficiently to dynamic market conditions and increasing the likelihood of achieving favorable outcomes in their own endeavors. Mantzari et al. (2017) defined mimetic pressures as occurring "when an organization attempts to imitate a more successful referent organization or improve upon the practice of other organizations". Boolaky et al. (2020) highlighted that "mimetic isomorphism arises from the replication of practices across nations, whereby there is a tendency to emulate what more successful countries have done to secure benefits and social acceptance".

Finally, normative isomorphism signifies the influence of universities and other professional organizations on firms, leading them toward homogeneity (Hassan 2008). DiMaggio and Powell (1997) further stressed that normative isomorphism is closely associated with a country's level of education. Hassan et al. (2014) emphasized that normative pressure resulting from the norms and values of the profession also influences the degree to which a nation will adopt international best practices. Boolaky et al. (2018) suggested that normative isomorphism occurs when individuals are trained under similar educational systems and tend to engage in similar conventional practices; they concluded that a firm that draws from a standard pool of professional staff would be able to improve its systems and practices because their ability to harmonize and enhance accounting quality may be greater.

The economic theory of networks suggests that countries are more inclined to adopt international standards, like IFRS, when they observe their economic partners already using them. According to Ramanna and Sletten (2009), IFRS is perceived as a commodity that countries have the discretion to embrace. The adoption decision is influenced by the network effect, wherein one country's adoption of IFRS encourages others to follow suit, leading to a network of countries utilizing the same standardized financial reporting framework. The decision to adopt international standards like IFRS is driven by two critical factors: the inherent value of the product and the network effects it creates, as described by Katz and Shapiro (1985).

Ramanna and Sletten (2009) put forward the idea that harmonizing accounting practices serves the purpose of globalizing trading networks. They introduced two key concepts: the "autarky value", which represents the inherent value of the product (accounting standards developed by the IASB), and the "synchronization value", which reflects the network value of the product arising from harmonization with other countries already using the same standards. According to the authors, a country should opt for international standards only when the combined benefits of both autarky and synchronization outweigh the advantages of sticking to local accounting standards.

#### 2.2. Information Disclosure and Cost of Equity Capital

Whether firms benefit from disclosure is one of the most critical issues in current accounting research. In particular, these benefits may arise from the reduced cost of equity capital brought about by companies' increased disclosure of accounting information. In recent years, several theoretical studies have focused on the relationship between the cost of equity capital and disclosure.

From a theoretical point of view, it has been argued that disclosure reduces information asymmetry and, consequently, the cost of equity capital for companies by reducing bid/ask spreads (Amihud and Mendelson 1986) or by increasing demand for a company's shares (Diamond and Verrecchia 1991). Another advantage of improving the quality of information is that it reduces the estimation risk of potential investors regarding the parameters of a stock's future performance. Indeed, investors are expected to assign greater systematic risk to poorly informed assets rather than highly informed ones (Clarkson et al. 1996).

Although many arguments favor accounting information quality and its positive impact on the cost of equity capital, theoretical discussions remain open. Thus, one of

the most controversial central questions in theoretical literature is whether the effects of information are diversified or not. Easley and O'Hara (2004) proposed a model of rational expectations in which information can influence a company's cost of equity capital, which is compatible with the logic of non-diversification. Indeed, a company can influence its cost of equity capital by acting on the accuracy and quantity of information made available to its investors. Furthermore, the authors believe that this objective can be achieved through a company's choice of accounting standards and disclosure policies.

In their study, Lambert et al. (2007) devised a methodology that establishes a connection between accounting information and the cost of equity capital. Their primary objective was to examine whether the quality of a company's accounting information is mirrored in the cost of its equity capital. Through this approach, the authors effectively demonstrated that the quality of accounting information has a dual impact on a company's cost of equity capital. Firstly, the quality of accounting information directly influences a company's cost of equity capital by shaping market players' perceptions of the distribution of future cash flows. Secondly, the quality of accounting information also has an indirect impact on a company's cost of equity capital through actual decisions made based on that information. Decisions taken by the company, which may alter the distribution of future cash flows, can further affect the cost of equity capital.

In several empirical studies, the relationship between information disclosure and information asymmetry/sharing costs varies according to the type of firm, type of disclosure, and measure of information asymmetry (Botosan 1997; Leuz and Verrecchia 2000; Botosan and Plumlee 2002; Francis et al. 2008). Furthermore, the effects of mandatory IFRS adoption on equity costs suggest that IFRS adoption can reduce equity costs in countries with strong enforcement and investor protection mechanisms (Daske et al. 2008; Li 2010; Persakis and Iatridis 2017).

In their research involving a sample of 307 Spanish-listed companies from 1999 to 2009, Castillo Merino et al. (2014) conducted a focused country-level analysis using OLS regression analysis. The dependent variable, the cost of equity capital, was estimated using the proxy proposed by Easton (2004). The authors discovered that Spanish-listed companies experienced a substantial decrease in their cost of equity capital following the compulsory adoption of IFRS in 2005. This reduction in the cost of equity capital remained significant even after accounting for various firm-specific risk factors and market-related variables that could potentially influence the cost of equity. Thus, increased financial disclosure, improved comparability of information, and changes in legal and institutional enforcement appear to have a joint effect on the cost of equity capital, leading to a sharp decrease in expected returns on equity.

Houge et al. (2016) conducted a study examining the impact of IFRS adoption on the cost of equity capital for listed companies in New Zealand. Their research was based on a sample of 290 firm-year observations spanning two periods: 1998–2002 and 2009–2013. The authors reported a significant negative association between IFRS adoption and the cost of equity capital, suggesting that IFRS is a higher-quality set of accounting standards than previous New Zealand GAAP. Their study provides empirical evidence on the impact of IFRS adoption on the cost of equity capital of New Zealand companies and supports the findings of previous studies on European companies.

In the case of Brazilian firms, Gatsios et al. (2016) assessed the impact of IFRS adoption on the cost of equity capital of 1325 Brazilian public companies over the period 2004–2013 using Difference-In-Difference (DID) analysis, which compares the results of firms that voluntarily adopted IFRS with those that adopted IFRS after the mandatory adoption period. Their results indicate that IFRS adoption did not reduce equity costs in Brazil. Similarly, Da Silva and Nardi (2017) studied the impact of IFRS adoption on Brazilian firms' cost of equity capital using DID and GMM approaches for 2010 and 2011. Their results show that an increase in information contributes to a reduction in asymmetric information and that a more efficient allocation of resources reduces the cost of equity capital. These results support the hypothesis of increased earnings quality after IFRS adoption.

Sanjaya et al. (2017) attempted to analyze and compare the cost of equity capital before and after the adoption of IFRS on the financial instrument of financial accounting standards (PSAK) for banking companies listed on the Indonesian stock exchange for the period 2008–2009 before IFRS adoption and 2013–2014 after IFRS adoption. The results of this study prove that the cost of equity capital was lower after IFRS adoption on financial instruments of financial accounting standards for banking companies listed on the Indonesian stock exchange. Thus, IFRS adoption reduces equity costs, impacts the reduction of non-performing loans, increases the loan-to-deposit ratio, and increases the net interest margin.

For a sample of 1658 firm-years from companies listed on the KSE and KOSDAQ from 2000 to 2013, Kim and Ryu (2018) studied the effect of mandatory IFRS adoption on the cost of equity capital, starting from its mandatory introduction in 2011 using the average implied cost of equity capital values presented by Claus and Thomas (2001), Gebhardt et al. (2001), Easton (2004), and Ohlson and Juettner-Nauroth (2005). Their results show a significantly negative relationship between mandatory IFRS adoption and the cost of equity capital, thus decreasing the cost of equity capital.

Not far away, De Moura et al. (2020) conducted a study to investigate the impact of mandatory IFRS adoption on the cost of equity capital and cost of debt for a group of firms operating in Argentina, Brazil, Chile, Mexico, and Peru. The findings reveal that even after controlling for firm-level reporting incentives, mandatory IFRS adoption reduces equity costs. Additionally, the cost of debt experienced a significant reduction after the IFRS adoption. These results suggest that the enhanced disclosure and comparability facilitated by IFRS standards, compared with previous domestic accounting standards, mitigated the information asymmetry problem and produced positive economic outcomes for firms operating in Latin America.

For their part, Saha and Bose (2021) examined the association between IFRS disclosure requirements and the cost of equity capital for a sample of 157 Australian firms. The authors showed that disclosure requirements negatively affect the cost of equity capital; thus, firms with higher IFRS disclosure levels have a lower cost of equity capital. Furthermore, the study revealed a negative relationship between IFRS disclosure requirements and the costs of debt and equity for the companies under investigation. These findings add valuable insights to the ongoing discussion about the comparative advantages and disadvantages of IFRS disclosure requirements. The implications of these results are significant for standard-setting bodies, regulators, and stakeholders who rely on financial statements for decision-making and analysis.

In a recent study, using a meta-analysis of 56 empirical studies with 1265 effect sizes, Opare et al. (2021) determined the impact of IFRS adoption on financial reporting comparability, market liquidity, cost of equity capital, and cost of debt. Their results show that IFRS adoption significantly improves comparability, increases market liquidity, and reduces the cost of equity capital but has no significant effect on the cost of debt. The results also show that mandatory IFRS adoption has a greater impact than voluntary adoption. However, for the cost of debt, voluntary adoption results in a reduction in the cost of debt but the impact of mandatory adoption on the cost of debt is not significant.

#### 2.3. Financial Instruments and Cost of Equity Capital

The risks associated with financial instruments are considered one of the most important aspects tested from the perspective of economic theory, along with the cost of capital. Despite the complexity of financial instruments, they are applied by all companies, including accounts receivable and payable as financial instruments that must be disclosed in every small or large company (Lim and Foo 2017). In addition, the introduction of financial instruments requires the disclosure of detailed information about the risks arising from the company's activities, such as liquidity risk, market risk, and credit risk (Jacobs 2009).

The importance of financial instruments in the implementation of IFRS and their different effects on the quality of financial reporting, investors, and capital markets have caused conflicts between researchers, accountants, and auditors. In addition, the fair value debate continues to be a controversial topic among academics in terms of its actual impact on the business domain, as fair value is at the core of financial instruments in IFRS implementation; thus, IFRS 7 brings the fair value debate to the forefront of disclosure requirements (Palea 2014; Kasyan et al. 2017). Moreover, IFRS 7 addresses the hedging policies used by companies in terms of cash flows, fair value, and foreign investments, as well as the relevant quantitative or qualitative information that investors and lenders consider important in assessing the situation of these companies (Deloitte 2017; Grosu and Chelba 2019).

According to Yamani et al. (2021), IFRS 7 financial instrument disclosures help to reduce information asymmetry. A better disclosure implies that companies adhere to the appropriate application of IFRS standards and meet their requirements. This shows that companies are committed to rules and regulations, thereby improving their level of transparency. Moreover, providing investors with comprehensive financial information on financial instruments enables companies to better understand their terms and conditions. This, in turn, can lead to a reduction in risk estimates and an improvement in capital market liquidity. As a result, investors and shareholders will benefit from greater confidence and closer relationships with companies, potentially leading them to demand a lower cost-of-capital ratio.

Financial intermediaries are generally very positive about IFRS standards when assessing potential borrowers. These standards promote transparency, consistency, and comparability, making it easier to make informed lending and risk assessment decisions, thus fostering a healthier financial ecosystem for both borrowers and lenders.

Balancing the benefits and costs of better-quality disclosure is crucial for companies. Striking the right balance can help businesses build trust with stakeholders, improve decision-making, and foster long-term sustainable growth while mitigating potential risks and resource burdens. Regulatory frameworks and industry standards play a critical role in guiding companies toward responsible and meaningful disclosure practices.

This framework has allowed us to deepen the complexities of disclosure practices and their implications. Taking into account both positive outcomes, such as increased transparency; better risk management and access to capital; and associated costs such as resource allocation, competitive disadvantage, and legal risks, this research can provide a more nuanced analysis of the subject.

# 2.4. Hypothesis Development

The relationship between mandatory IFRS disclosures and the cost of equity capital has been neglected, despite its potential significance in the disclosure overload problem debate. Some studies have examined the impact of IFRS disclosure on firms' cost of equity capital and are essential for providing additional information and clarifying firms' accounting policies and calculations. However, there needs to be more research on the association between mandatory IFRS disclosure and the cost of equity capital, particularly in the context of the disclosure overload debate. Disclosure under the various IFRS measurement and recognition requirements should help reduce the cost of equity capital. Thus, based on this reasoning, we propose the following hypothesis:

# **H1.** The level of IFRS disclosure reduces companies' cost of equity capital exposure.

In other words, the more a company discloses under IFRS, the lower its cost of equity capital. This hypothesis can be tested by the collection of data on a sample of firms and by analyzing the relationship between the cost of equity capital and the level of IFRS disclosure. It is important to note that proving causality between two variables is only sometimes possible and other factors may influence the results.

#### 3. Methodology

## 3.1. Sample and Data

As the mandatory transition to IFRS has concerned listed companies located in the European Union, we followed Ertz et al. (2021) by testing the effect of IFRS on the cost of equity capital by considering 337 firms listed on the STOXX Europe 600 over the period 1994–2022, i.e., a total of 9773 firm-year observations. This stock market index includes the 600 largest market capitalizations in 17 European countries: Austria, Belgium, Denmark, Finland, France, Germany, Ireland, Italy, Luxembourg, the Netherlands, Norway, Poland, Portugal, Spain, Sweden, Switzerland, and the United Kingdom. This choice is motivated by the idea that, although each company has been affected differently by the transition to IFRS, these impacts are homogeneous within a single industry. Therefore, we consider seven industries represented in STOXX Europe 600 with different characteristics. Following Lotfi et al. (2022, 2023), the selected industries were automotive, healthcare, food and beverage, and banking.

The companies selected were all listed on STOXX Europe 600 when they published their financial statements under IFRS Standards, mainly in 2004 or 2003. As the impact of IFRS may differ depending on the sector of activity and the environment in which the company operates, it is crucial to consider this in our analysis and diversify the countries where the companies were headquartered at the time of this accounting transition as much as possible. Information on the selected companies by industry sector and head office country is summarized in Table 1.

Number of Sectors	Sector of Activity	Number of Firms	Percentage
1	Consumer goods	35	10.39%
2	Technology	27	8.01%
3	Health	49	14.54%
4	Oil and Gas	22	6.53%
5	Industry	154	45.70%
6	Telecommunications	18	5.34%
7	Consumer Services	32	9.50%
	Total	337	100%

**Table 1.** Distribution of the final sample by sector of activity.

## 3.2. Cost of Equity Capital Measure

The variable chosen for the statistical analysis is the cost of equity capital, defined as the opportunity cost that evaluates investors' interest in investing their money in a company rather than elsewhere. It represents the minimum rate of return that must be generated by the company's investments in order for it to meet the profitability requirements of shareholders and creditors. Therefore, to estimate the cost of equity capital, referring to Houqe et al. (2016), we use the modified Price–Earnings–Growth (PEG) ratio model proposed by Easton (2004). Modification of the standard PEG ratio model involves inclusion in the model of a dividend per share forecast one year in advance. Botosan and Plumlee (2005) conclude that estimates of the modified PEG ratio model provide the best measure of the cost of equity capital in a country with strong investor protection because it dominates the other alternatives in that it is consistently and predictably linked to various risk measures such as information risk, leverage risk, residual risk, market risk, and growth. Thus, given strong investor protection, we use the modified PEG ratio model as follows:

$$K_{e} = \frac{eps_{t+2} - eps_{t+1} + K_{e} * Div_{t+1}}{P_{t}} \text{ or } K_{e} = \frac{eps_{t+2} - eps_{t+1}}{P_{t} - Div_{t+1}}$$
 (1)

where  $K_e$  is the cost of equity capital,  $eps_{t+1}$  is the expected earnings per share at the one-year horizon,  $eps_{t+2}$  is the expected earnings per share at the two-year horizon,  $Div_{t+1}$  is the one-year-ahead dividend forecast, and  $P_t$  is the price per share at year-end.

## 3.3. Estimation Technique

To test the effect of IFRS adoption on the cost of equity capital for 337 firms from 17 European countries between 1994 and 2022 chosen from STOXX Europe 600-listed companies, we adopt the following regression equation, which includes a set of company-specific controls for other factors that may affect a company's cost of equity capital. We use the IFRS variable, which indicates the change in the accounting framework following the mandatory adoption of IFRS in Europe since 2005; it takes 0 before the mandatory adoption of IFRS in 2005 and 1 after the mandatory adoption of IFRS. Concerning Houge et al. (2016), and the GMM-system suggested by Arellano and Bover (1995) as well as Blundell and Bond (1998), the model can be written as follows:

$$Ke_{it} = \beta_0 + \beta_1 Ke_{it-1} + \beta_2 IFRS_{it} + \beta_3 Size_{it} + \beta_4 BMR_{it} + \beta_5 Beta_{it} + \beta_6 FL_{it} + \beta_7 ROE_{it} + \epsilon_{it}$$
 (2)

Let  $Ke_{it}$  represent the cost of equity capital for firm "i" i in year "t". Additionally, let IFRS be a dichotomous variable that takes the value of 1 when the financial statements of the firm "i" are prepared in accordance with IFRS in a year "t", and 0 otherwise. Size is measured by the natural logarithm of the current year's total assets for a firm "i" in a year "t". BMR is the ratio of the market value of equity to book value of equity for a firm "i" in a year "t". Beta is the systematic risk of firm i in year t. FL represents the firm's financial leverage, which is the ratio of total Debt to Shareholders' Equity of a firm "i" in a year "t". ROE is the return on equity, which measures financial performance and is calculated by dividing net income by shareholders' equity of a firm "i" in a year "t".  $\varepsilon_{it}$  is an error term assumed to verify the statistical properties of white noise regardless of a firm "i" or a period "t". We summarize all the variables in Table 2.

**Table 2.** Variable description.

Variables	Definition
Ke	Cost of Equity Capital
IFRS	The dichotomous variable that is equal to 1 when the financial statements are prepared in accordance with IFRS and 0 otherwise
Size	Measured by the natural logarithm of the current year's total assets for firm i in year t
BMR	The ratio of the market value of equity to the book value of equity for firm i in year t
Beta	The systematic risk of firm i in year t
FL	The ratio of total Debt to Shareholders' Equity of firm i in year t
ROE	The return on equity, which measures financial performance and is calculated by dividing net income by the shareholders' equity of firm i in year t

To address potential bias and inaccuracies associated with using difference GMM (Arellano and Bond 1991), Arellano and Bover (1995) as well as Blundell and Bond (1998) propose a system of difference and level regressions. In the difference regression, the instruments are the lagged levels of the explanatory variables, while in the level regression, the instruments are the lagged differences of the explanatory variables. These instruments are considered appropriate under the assumption that while there might be a correlation between the levels of the explanatory variables and the country-specific effect, there is no correlation between these variables in the differences and country-specific effects.

The consistency of the GMM-system estimator relies on two key aspects: the validity of the assumption that the error term is serially uncorrelated and the validity of the instruments. The test of the null hypothesis of no first-order serial correlation should be rejected under the identification assumption that the error is serially uncorrelated, whereas the test of the null hypothesis of no second-order serial correlation should not be rejected. Therefore, to evaluate the model's performance and instrument validity, we employ two diagnostic tests proposed by Arellano and Bover (1995) and by Blundell and Bond (1998). Additionally, we use the Hansen (1982) tests of over-identifying restrictions; if the null

hypothesis cannot be rejected, it would indicate that the model is correctly specified and the instruments are valid.

## 4. Empirical Results

#### 4.1. Descriptive Statistics

Before commencing the examination of variables' stationarity, cointegration relationship, cross-sectional dependence analysis, and model analysis, it is crucial to initiate the process with a descriptive and graphical analysis. This preliminary analysis will serve as the foundation for subsequent estimations and assessments.

According to the information presented in Table 3, the variable "Ke" exhibits the following descriptive statistics: The overall mean of the variable is 0.014, with a low median value of 0.005. The standard deviation is 0.719, and the minimum and maximum values are -37.303 and 37.602, respectively. The distribution of the variable is highly left-skewed, as indicated by the skewness value of -7.179, which is less than 0. Additionally, the distribution is strongly platykurtic, with a kurtosis value of 2079.617, which exceeds 0, signifying heavy tails and extreme outliers. The dataset comprises a total of 9773 observations. It is important to note that the distribution of the variable "Ke" is non-normal for the entire sample and demonstrates no autocorrelation. In addition, the fact that the median is low (0.5%) proves once again that the distribution is asymmetrical and there is a strong asymmetry of information concerning this variable Ke.

**Table 3.** Descriptive statistics of the variables in the sample.

Variables	Ke	IFRS	Size	BMR	Beta	FL	ROE
Observations	9773	9773	9773	9773	9773	9773	9773
Mean	0.014	0.621	15.598	3.231	0.893	0.583	19.436
Standard deviation	0.719	0.485	1.990	10.008	0.973	0.205	76.295
Minimum	-37.303	0	8.301	-548.090	-19.069	0.005	-3043.680
Maximum	37.602	1	21.010	204.570	8.322	2.693	2230.020
Median	0.005	1	15.750	2.380	0.880	0.587	15
Skewness	-7.179	-0.497	-0.393	-25.591	-8.855	1.301	2.374
Kurtosis	2079.617	1.247	2.949	1395.255	160.138	13.500	521.683
Jarque–Bera (JB) test	$1.8 \times 10^{9}$	-	252.6	$7.9 \times 10^8$	$1.0 \times 10^{7}$	$4.8  imes 10^4$	$1.1 \times 10^{8}$
Probability JB	0.000	-	0.000	0.000	0	0	0
Born-Breitung (BB) test	2.300	-	225.330	0.240	15.890	59.500	4.690
Probability BB	0.317	-	0.000	0.889	0.000	0.000	0.096

Notes: BB refers to Born and Breitung's (2016) serial correlation test. JB refers to Jarque and Bera's (1987) normality test.

According to the data presented, for the 9773 observations, the variable "IFRS" is described by the following statistics: The overall mean of the variable is 0.621 and the median value is 1. The standard deviation is 0.485, and the minimum and maximum values of the variable are 0 and 1, respectively. The distribution of the variable "IFRS" is highly left-skewed, as evident from the negative skewness value of -0.497, which is less than 0. Moreover, the distribution is leptokurtic, with a kurtosis value of 1.247, which exceeds 0, indicating heavy tails and more extreme values.

After global descriptive statistical interpretation, we first performed a unit root test for the variables of the model. In this step, we first test the null hypothesis of cross-sectional independence between individuals. De Hoyos and Sarafidis (2006) emphasize the need and significance of conducting a cross-sectional dependence test when working with dynamic panel data. In particular, Sarafidis and Robertson (2006) underscore that the presence of cross-sectional dependence in the data is crucial to avoid inconsistencies in all estimation procedures. Hence, in this study, we explore various dependence tests to ensure the reliability of our analysis, as cited in Pesaran (2021). The p-values associated with the different CD tests are below 0.05, suggesting that augmentation with current and lagged cross-sectional averages adequately accounts for cross-sectional dependence (see Table 4).

Table 4.	Cross-section	depend	lency tests.

Tests	Value	Probability	Decision
Friedman (1937)	888.479	0.000	Dependence
Frees (1995, 2004)	6.033	0.000	Dependence
Pesaran (2006)	89.162	0.000	Dependence
Pesaran (2015)	103.813	0.000	Dependence

Second, after performing the cross-dependence tests cited by Pesaran (2021), we examine the unit root tests for the model variables. In this step, we examine the unit root tests by two generations; the first generation is represented by Levin et al. (2002), Im et al. (2003), as well as Hadri (2000), while the second is represented by Pesaran (2003) and Pesaran (2007) unit root tests.

First-generation unit root tests are based on the assumption that the residuals are interindividually independent. This assumption allows for the straightforward establishment of statistical distributions for tests, often resulting in asymptotic or semi-asymptotic normal distributions. In contrast, second-generation unit root tests typically depart from the independence assumption. These tests adopt a completely different perspective in which correlations between individuals are not considered nuisance parameters. Instead, they propose leveraging these co-movements to define new test statistics.

According to first-generation unit root tests conducted by Levin et al. (2002), Im et al. (2003), and Hadri (2000) presented in Table 5, the variables in the model are either level stationary or first difference stationary for all variables in the model. However, for the second-generation tests of Pesaran (2003) and Pesaran (2007) presented in Table 6, all variables are stationary in the first difference.

**Table 5.** The first generation of unit root tests.

Variables		In Level		In	First Differer	ice
variables	LLC	IPS	Hadri	LLC	IPS	Hadri
Ke	-54.564 ***	-58.486 ***	15.152 ***	-88.892 ***	-70.382 ***	-18.036 ***
Size	-13.488***	6.497 ***	280.411 ***	-36.259 ***	-46.823***	9.783 ***
BMR	-6.273***	-13.312***	3.787 ***	-47.688***	-57.280 ***	-18.237 ***
Beta	-12.694 ***	-5.013 ***	184.773 ***	-38.816 ***	-46.982***	1.178 ***
FL	-11.498 ***	-10.209 ***	160.231 ***	-47.375***	-53.455 ***	-2.446***
ROE	-12.678 ***	-25.124 ***	92.294 ***	-47.286 ***	-59.689 ***	-8.883 ***

Note: \*\*\* represent significance at 1%.

We use the unit root test with breaks suggested by Karavias and Tzavalis (2014) to verify the unit root tests mentioned above. The results in Table 7 show that the series is stationary in level or first difference related to certain breaks in 1995, 1997, 2000, and 2021, related to the European Monetary System crisis (1992–1993), Asian Financial crisis (1997–1998), Internet bubble crisis (2001), and COVID-19 crisis (2019–2020), respectively. Therefore, it is necessary to check for the existence of a cointegrating relationship between the series.

Given that the majority of variables exhibit stationarity when analyzed in their first difference, it becomes crucial to investigate whether a cointegrating relationship exists among these variables. Granger (1981) showed that when a series is integrated in order one (they become stationary after the first differencing) but their linear combination is already stationary without differencing, they are said to be cointegrated, which implies the existence of a long-run relationship between the series (Mahmoodi and Mahmoodi 2016). Based on the outcomes presented in Table 8, which include various cointegration tests like those by Kao (1999) and Pedroni (2004), the results indicate that the probability obtained from both tests falls below the 5% significance threshold. As a result, we can infer that there is at least one cointegrating relationship among all the variables included in our model.

**Table 6.** Second-generation unit root tests.

Variables	Ke	Size	BMR	Beta	FL	ROE	
		Pesaran (2003)	) unit root test				
		Panel A:	In level				
Constant	-4.350 ***	-1.821	-1.787	-2.263 ***	-2.067 ***	-2.401 ***	
Constant and Trend	-4.458***	-2.204	-2.331	-2.295	-2.304	-2.678***	
Decision	S	NS	NS	NS	NS	S	
		Panel B: In fi	rst difference				
Constant	-5.661 ***	-3.403 ***	-4.103 ***	-3.210 ***	-3.716 ***	-4.356 ***	
Constant and Trend	-5.692 ***	-3.568 ***	-4.233***	-3.338 ***	-3.847***	-4.399 ***	
Decision	S	S	S	S	S	S	
		Pesaran (2007)	) unit root test				
		Panel A:	In level				
Constant	-5.427 ***	-1.949	-2.282 ***	-2.266 ***	-2.221 ***	-3.023 ***	
Constant and Trend	-5.586 ***	-2.400***	-2.868***	-2.318 ***	-2.474***	-3.362 ***	
Decision	S	NS	S	S	S	S	
Panel B: In first difference							
Constant	-6.085 ***	-4.795 ***	-5.606 ***	-4.432 ***	-5.092 ***	-5.589 ***	
Constant and Trend	-6.265 ***	-4.946***	-5.761 ***	-4.543 ***	-5.216 ***	-5.734 ***	
Decision	S	S	S	S	S	S	

Note: \*\*\* represent significance at 1%.

**Table 7.** Unit root test with break.

Variables	In Level	In First Difference
Ke	$-1.9 \times 10^2 *** (1995)$	$-2.4 \times 10^2$ *** (2021)
Size	-85.779 *** (2021)	$-1.4 \times 10^2 *** (2021)$
BMR	$-1.5 \times 10^2$ *** (2021)	$-2.3 \times 10^2 *** (2021)$
Beta	-25.416 *** (2000)	$-1.5 \times 10^2 *** (2021)$
FL	-23.654 *** (1997)	$-1.6 \times 10^2$ *** (2021)
ROE	-53.352 *** (2021)	$-1.8 \times 10^2 *** (2021)$

Note: \*\*\* represent significance at 1%.

Table 8. Cointegration tests.

Tests	t-Statistic	Probability	Decision
Kao (1999)	-60.573	0.000	Cointegration
Pedroni (2004)	-107.764	0.000	Cointegration

#### 4.2. Estimation and Interpretation

After examining the stationarity and cointegration tests, a set of robustness tests were performed. In fact, the results show that our panel is characterized by a serial autocorrelation problem ( ${\rm chi}^2$  (136) = 1712.999, p-value = 0.000), a heteroscedasticity problem ( ${\rm chi}^2$  (17) = 138.68, p-value = 0.000), and presents a cross-sectional dependency problem (Table 4). To address these issues effectively, the "Robust" command in Stata was utilized in conjunction with the Generalized Method of Moments (GMM) system approach proposed by Arellano and Bover (1995) and Blundell and Bond (1998).

In addition, we examine diagnostic tests such as the Arellano and Bond (1991) autocorrelation test and the over-identification tests of Hansen (1982) to validate the estimation of this model.

The estimation established in Table 9 shows that the cost of equity capital decreases after IFRS adoption. In fact, the adoption reduces the cost of equity capital by 0.038, which is consistent with the findings of Prather-Kinsey et al. (2008), Daske et al. (2008), Armstrong et al. (2010), Li (2010), Palea (2013), Castillo Merino et al. (2014), Houqe et al. (2016), Persakis and Iatridis (2017), Utama et al. (2017), Wook-Bin and Yuk (2018), and De Moura

et al. (2020), who found that the IFRS Standards reduce the cost of equity capital. IFRS adoption has been beneficial to EU capital markets because it is associated with greater earnings and equity value relevance, increased information content, and lower cost of equity capital than before IFRS adoption. In addition, it improves the comparability of financial statements, which enhances the ability of users of financial statements to assess business performance. The improvement in disclosure and comparability of financial statements also reduces the cost of equity capital.

Table 7. I WO Step Civilyi System Csimilation	Table 9.	. Two-step	GMM-system	estimation
---	----------	------------	------------	------------

Variables	Coefficient	Corrected Standard-Deviation	t-Statistic	Probability			
Ke <sub>it-1</sub>	-0.106	0.004	-28.92	0.000			
IFRS <sub>it</sub>	-0.038	0.012	-3.17	0.002			
Size <sub>it</sub>	0.014	0.004	3.21	0.001			
$BMR_{it}$	0.0001	0.0002	0.52	0.603			
Beta <sub>it</sub>	0.006	0.003	1.78	0.077			
$FL_{it}$	0.081	0.032	2.51	0.013			
$ROE_{it}$	-0.001	0.0001	-2.68	0.008			
Constant	-0.226	0.065	-3.46	0.001			
AR(1) test	-1.42 (0.155)						
AR(2) test		-0.04(0.967)					
Hansen test		23.510 (0.133	)				

The regression analysis of European firms reveals that among the control variables, size (firm size) has a significant and positive coefficient (p-value = 0.001 < 1%), indicating that it has a significant effect on the increase in the cost of equity capital. This finding is consistent with the results of Reschiwati et al. (2020) and is explained by trade-off theory. According to this theory, larger firms can use more debt because they have lower bankruptcy risk. The bankruptcy risk for large firms translates into a lower cost of debt, which encourages them to use more debt. Therefore, firm size plays a crucial role in the cost of equity capital.

As expected, systematic risk (beta) had a positive and significant effect at the 10% level (p-value = 0.077 < 10%). This reveals that, as systemic risk increases, firms' cost of equity capital increases. Our results are consistent with those of Castillo Merino et al. (2014) and Houqe et al. (2016). Financial leverage (FL) also has a positive and significant effect at the 5% threshold (p-value = 0.013 < 5%) on the cost of equity capital in EU countries. This result was confirmed by Castillo Merino et al. (2014) and Persakis and Iatridis (2017).

This discount rate is an important element of corporate financial policy and influences the performance of capital markets because a slight variation in its value significantly affects the stock market value of a firm and its ability to create value. As expected, a firm's beta and leverage significantly and positively affect the cost of equity capital. Sharpe (1964) assumes that the higher the risk of a stock, the higher the return expected by investors. The estimation risk argument also affects leverage because a higher level of leverage increases the estimation risk of potential investors regarding the parameters of a stock's future returns.

The result for ROE shows a negative and significant relationship at the 1% level with the cost of equity capital in the EU countries (p-value = 0.008 < 1%). This result is confirmed by Ali Shah and Butt (2009), Khan (2016), and Faysal et al. (2021). Return on equity (ROE) (net income after tax/equity) measures a firm's return on equity. It was used as a control variable in the research on the relationship because of its impact on firm risk. Thus, the higher the ROE, the more comfortable investors are and the lower the risk. In theory, ROE is a profitability ratio that measures a company's ability to manage its sources of funds to increase revenues. If the ROE generated is high, it means that management has been able to

manage the existing capital as much as possible so that the profit generated is high, which should attract investors to invest in the company.

The diagnostic tests of the GMM-system method, as displayed in Table 9, indicate that the Arellano and Bond (1991) tests for AR(1) and AR(2) are not statistically significant. In addition, Hansen's (1982) test shows that the instruments are identified. Thus, these two tests for the GMM system estimation confirm the validity of this estimation.

European countries were early adopters of IFRS, especially after the European Union (EU) made it mandatory for listed companies to prepare their consolidated financial statements in accordance with IFRS since 2005. Many non-listed companies also choose to apply IFRS for their financial reporting due to the benefits of consistency, comparability, and global acceptance.

As a result of widespread IFRS adoption in Europe, investors and analysts are more familiar with the IFRS financial statements and reporting standards. This familiarity can lead to greater transparency and understanding of financial information, potentially reducing information asymmetry between companies and investors. Consequently, this can result in a reduction in the perceived risk by investors, leading to a lower cost of equity for European companies.

Non-European countries have also made significant progress in adopting IFRS; however, the extent of its adoption varies. Some countries have fully adopted IFRS for both listed and non-listed companies, while others may have adopted it only partially or for specific industries. In countries where IFRS adoption is limited, investors may face challenges in understanding and analyzing financial statements prepared using local accounting standards, especially if they are unfamiliar with those standards. This could result in increased uncertainty and perceived risk for investors, leading to a higher cost of equity for companies.

## 5. Conclusions and Policy Implications

This study examines the impact of IFRS adoption on the cost of equity capital of 337 firms listed in STOXX Europe 600, spanning 17 European countries, from 1994 to 2022. To estimate the cost of equity capital, the modified price–earnings–growth ratio model was employed, and the GMM-system technique suggested by Arellano and Bover (1995) and Blundell and Bond (1998) was used.

The findings reveal that IFRS adoption is negatively associated with European firms' cost of equity capital. In other words, IFRS adoption leads to a 0.038 reduction in the cost of equity capital. This implies that firms in European countries have benefited from a decrease in the cost of equity capital after IFRS adoption. This result supports the argument that high-quality accounting standards enhance the quality of financial reporting, which could positively affect firms' cost of equity capital, provided that adopting new accounting standards are implemented with a focus on improving a country's enforcement mechanisms. Thus, this result supports the hypothesis above.

After conducting a regression analysis, linking the estimated cost of equity capital of European firms to various control variables concerning accounting information, market influence, and mandatory IFRS adoption, we find compelling evidence regarding the impact of risk parameters (beta) on expected stock returns. Specifically, a firm's beta has a significant and positive effect on its cost of equity capital. An increase in a company's beta leads to a rise in its cost of equity capital, indicating higher perceived risk for investors and, consequently, a demand for increased returns to invest in the firm's stocks. Moreover, the study reveals that firms with higher leverage tend to possess a riskier profile, which leads investors to seek higher returns when investing in their stocks. The evidence consistently supports the idea that leverage not only positively influences the return on equity but also significantly affects the cost of equity capital for firms. This effect is attributed to the increase in the discount rate applied to future cash flows, thereby reducing the value of equity for investors.

After controlling for market beta and leverage, we find that—in contrast to Daske et al. (2008) and similar to the results of Li (2010), Castillo Merino et al. (2014), Houqe et al. (2016), and Persakis and Iatridis (2017)—the mandatory adoption of IFRS by European firms in 2005 led to a decrease in the cost of equity capital. Thus, improved financial disclosure quality and enforcement mechanisms have a significant and negative joint effect on the cost of equity capital of European stocks. This finding is relevant, as it suggests that a country-specific analysis with additional data for the post-adoption period is needed to capture and understand the economic consequences of mandatory IFRS adoption at the national level.

Thus, IFRS standards should be encouraged. Two key policy implications must be considered. First, European countries have to invest in Education and Training. Since IFRS adoption requires significant knowledge and expertise, the EU should invest in education and training programs to improve accounting professionals, investors, and other stakeholders' understanding and application of IFRS. Policymakers can prioritize developing high-quality training programs that cover the latest updates and changes in IFRS standards. By doing so, the EU can ensure that accounting professionals have the skills to implement IFRS effectively.

Second, European countries can harmonize their tax laws. In fact, the differences between EU countries' tax laws can create challenges in IFRS implementation. Policymakers can harmonize tax laws to ensure consistency and reduce the compliance burden on businesses. Harmonizing tax laws can also reduce the potential for tax-related distortions in financial reporting. This can increase investor confidence in financial statements and improve the comparability of financial information across the EU.

As in any research, our study has several perspectives. First, our findings rely on estimating the effect of IFRS disclosures in European countries. Future research could extend our study to other regions or countries, such as Asia and Africa. Second, our results show that IFRS adoption may be the only way to affect a firm's cost of equity capital and that there are many other potential factors in the literature that may have a larger impact than IFRS adoption, such as financial instruments and corporate governance. Third, future studies could explore how IFRS adoption affects equity costs in different economic sectors. Finally, further research on the implications of IFRS can be expanded and differentiated based on Europe countries, company dimensions, or business sectors. This approach would allow for a more comprehensive understanding of how IFRS impacts accounting and finance practices across different geographical regions, company sizes, and industry sectors.

**Author Contributions:** All authors contributed equally to the work. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

**Institutional Review Board Statement:** Not applicable.

**Informed Consent Statement:** Title of product: Do IFRS Disclosure Requirements Reduce the Cost of Equity Capital? Evidence from European Firms. Main authors: Ghouma Ghouma, Hamdi Becha, Maha Kalai, Kamel Helali. (Affiliation: Faculty of Economics and Management of Sfax, University of Sfax, Tunisia) and Myriam Ertz (Affiliation: Department of Economics and Administrative Sciences, University of Quebec at Chicoutimi, Canada). This is to state that I give my full permission for the publication and reproduction in all editions of the above-named product. I hereby agree to release and discharge any editors or other contributors and their agents, publishers, successors and assigns from any and all claims, demands or causes of action that I may now have or may hereafter have for libel, defamation, invasion of privacy, copyright or moral rights or violation of any other rights arising out of or relating to any use of my image or case history.

**Data Availability Statement:** The data that support the findings of this study are available from DataStream databases. Data are however available from the authors upon request. We use Stata 17.0 software to run different programs. Codes are available on request.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- AICPA. 2005. International financial reporting standards international financial reporting standards. *International Financial Reporting Standards* 44: 26–28.
- Ali Shah, Syed Zulfiqar, and Safdar Ali Butt. 2009. The impact of corporate governance on the cost of equity: Empirical evidence from Pakistani listed companies. *The Lahore Journal of Economics* 14: 139–71. [CrossRef]
- Ali, Syed Tauseef, Zhen Yang, Zahid Sarwar, and Farman Ali. 2019. The impact of corporate governance on the cost of equity: Evidence from cement sector of Pakistan. *Asian Journal of Accounting Research* 4: 293–314. [CrossRef]
- Amihud, Yakov, and Haim Mendelson. 1986. Asset Pricing and the Bid-Ask Spread. *Journal of Financial Economics* 17: 223–49. [CrossRef] Arellano, Manuel, and Olympia Bover. 1995. Another look at the instrumental variable estimation of error-components models. *Journal of Econometrics* 68: 29–51. [CrossRef]
- Arellano, Manuel, and Stephen Bond. 1991. Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *The Review of Economic Studies* 58: 277–97. [CrossRef]
- Armstrong, Christopher S., Mary E. Barth, Alan D. Jagolinzer, and Edward J. Riedl. 2010. Market reaction to the adoption of IFRS in Europe. *The Accounting Review* 85: 31–61. [CrossRef]
- Ball, Ray. 2006. International Financial Reporting Standards (IFRS): Pros and cons for investors. *Accounting and Business Research* 36: 5–27. [CrossRef]
- Berle, Adolf, and Gardner Means. 1932. The Modern Corporation and Private Property. New York: Macmillan Press.
- Blundell, Richard, and Steve Bond. 1998. Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics* 87: 115–43. [CrossRef]
- Boolaky, Pran K., Kamil Omoteso, Masud U. Ibrahim, and Ismail Adelopo. 2018. The development of accounting practices and the adoption of IFRS in selected MENA countries. *Journal of Accounting in Emerging Economies* 8: 327–51. [CrossRef]
- Boolaky, Pran K., Vincent Tawiah, and Teerooven Soobaroyen. 2020. Why do African countries adopt IFRS? An institutional perspective. *The International Journal of Accounting* 55: 2050005. [CrossRef]
- Born, Benjamin, and Jörg Breitung. 2016. Testing for serial correlation in fixed-effects panel data models. *Econometric Reviews* 35: 1290–316. [CrossRef]
- Botosan, Christine A. 1997. Disclosure level and the cost of equity capital. Accounting Review 72: 323-49.
- Botosan, Christine A., and Marlene A. Plumlee. 2002. A re-examination of disclosure level and the expected cost of equity capital. *Journal of Accounting Research* 40: 21–40. [CrossRef]
- Botosan, Christine A., and Marlene A. Plumlee. 2005. Assessing alternative proxies for the expected risk premium. *The Accounting Review* 80: 21–53. [CrossRef]
- Castillo Merino, David, C. Menéndez-Plans, and Neus Orgaz-Guerrero. 2014. Mandatory IFRS adoption and the cost of equity capital: Evidence from Spanish firms. *Intangible Capital* 10: 562–83. [CrossRef]
- Clarkson, Pete, Jose Guedes, and Rex Thompson. 1996. On the diversification, observability, and measurement of estimation risk. *Journal of Financial and Quantitative Analysis* 31: 69–84. [CrossRef]
- Claus, James, and Jaco Thomas. 2001. Equity premia as low as three percent? Evidence from analysts' earnings forecasts for domestic and international stock markets. *The Journal of Finance* 56: 1629–66. [CrossRef]
- Da Silva, Ricardo Luiz Menezes, and Paula Carolina Ciampaglia Nardi. 2017. Full adoption of IFRSs in Brazil: Earnings quality and the cost of equity capital. *Research in International Business and Finance* 42: 1057–73. [CrossRef]
- Damak-Ayadi, Salma, Nesrine Sassi, and Moujib Bahri. 2020. Cross-country determinants of IFRS for SMEs adoption. *Journal of Financial Reporting and Accounting* 18: 147–68. [CrossRef]
- Daske, Holge, Luzi Hail, Christian Leuz, and Rodrigo Verdi. 2008. Mandatory IFRS reporting around the world: Early evidence on the economic consequences. *Journal of Accounting Research* 46: 1085–142.
- De Fond, Mark, Xuesong Hu, Mingyi Hung, and Siqi Li. 2011. The impact of mandatory IFRS adoption on foreign mutual fund ownership: The role of comparability. *Journal of Accounting and Economics* 51: 240–58. [CrossRef]
- De Hoyos, Rafael E., and Vasilis Sarafidis. 2006. Testing for cross-sectional dependence in panel-data models. *The Stata Journal* 6: 482–96. [CrossRef]
- De Moura, André Aroldo Freitas, André Altuwaijri, and Jairaj Gupta. 2020. Did mandatory IFRS adoption affect the cost of capital in Latin American countries? *Journal of International Accounting, Auditing and Taxation* 38: 100301. [CrossRef]
- Déjean, Frédérique, and Michèle Saboly. 2006. *La Profession Comptable Française: Entre Mimétisme et Rivalité?* Cahiers de Recherche, 175. Toulouse: IAE de Toulouse. Available online: https://www.researchgate.net/publication/255625962\_La\_profession\_comptable\_française\_entre\_mimetisme\_et\_rivalite (accessed on 7 August 2023).
- Deloitte. 2017. IFRS 7: Financial Instruments: Disclosures. Available online: https://www.iasplus.com/en-gb/standards/ifrs-en-gb/ifrs7 (accessed on 3 April 2017).
- Diamond, Douglas W., and Robert E. Verrecchia. 1991. Disclosure, liquidity, and the cost of capital. *The Journal of Finance* 46: 1325–59. [CrossRef]
- DiMaggio, Paul, and Walter Powell. 1983. The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review* 48: 147–60. [CrossRef]
- DiMaggio, Paul, and Walter Powell. 1997. Le néo-institutionnalisme dans l'analyse des organisations. *Politix. Revue des Sciences Sociales du Politique* 10: 113–54.

Easley, David, and Maureen O'Hara. 2004. Information and the cost of capital. The Journal of Finance 59: 1553–83. [CrossRef]

Easton, Peter D. 2004. PE ratios, PEG ratios, and estimating the implied expected rate of return on equity capital. *The Accounting Review* 79: 73–94. [CrossRef]

Eisenhardt, Kathleen M. 1989. Agency theory: An assessment and review. Academy of Management Review 14: 57-74. [CrossRef]

Ertz, Myriam, Shouheng Sun, and Imen Latrous. 2021. The impact of big data on firm performance. In *Advances in Digital Science: ICADS 2021.* Edited by Tatiana Antipova. Berlin: Springer International Publishing, pp. 451–62.

Fama, Eugene F., and Michael C. Jensen. 1983. Separation of ownership and control. The Journal of Law & Economics 26: 301–25.

Faysal, Saad, Mahdi Salehi, and Mahdi Moradi. 2021. Impact of corporate governance mechanisms on the cost of equity capital in emerging markets. *Journal of Public Affairs* 21: e2166. [CrossRef]

Francis, Jennifer, Dhanajay Nanda, and Per Olsson. 2008. Voluntary disclosure, earnings quality, and cost of capital. *Journal of Accounting Research* 46: 53–99. [CrossRef]

Frees, Edward W. 1995. Assessing cross-sectional correlation in panel data. Journal of Econometrics 69: 393-414. [CrossRef]

Frees, Edward W. 2004. Longitudinal and Panel Data: Analysis and Applications in the Social Sciences. Cambridge: Cambridge University Press.

Friedman, Milton. 1937. The use of ranks to avoid the assumption of normality implicit in the analysis of variance. *Journal of the American Statistical Association* 32: 675–701. [CrossRef]

Gatsios, Rafael Confetti, Josè Marcos Da Silva, Marcelo Augusto Ambrozini, Alexandre Assaf Neto, and Fabiano Guasti Lima. 2016. Impact of adopting IFRS standard on the equity cost of Brazilian open capital companies. *RAM. Revista de Administração Mackenzie* 17: 85–108. [CrossRef]

Gebhardt, William R., Charles M. Lee, and Bhaskaran Swaminathan. 2001. Toward an implied cost of capital. *Journal of Accounting Research* 39: 135–76. [CrossRef]

Granger, Clive W. 1981. Some properties of time series data and their use in econometric model specification. *Journal of Econometrics* 16: 121–30. [CrossRef]

Grosu, Veronica, and Ancuta Anisia Chelba. 2019. IFRS 7-Financial Instruments Disclosures. Ecoforum Journal 8: 1-6.

Hadri, Kaddour. 2000. Testing for stationarity in heterogeneous panel data. The Econometrics Journal 3: 148-61. [CrossRef]

Han, Bing, Ya Tang, and Liyan Yang. 2016. Public information and uninformed trading: Implications for market liquidity and price efficiency. *Journal of Economic Theory* 163: 604–43.

Hassan, Enas A., Michaela Rankin, and Wei Lu. 2014. The development of accounting and reporting in Iraq and the IFRS decision: An institutional perspective. *The International Journal of Accounting* 49: 371–90. [CrossRef]

Hassan, Mostafa Kamal. 2008. The development of accounting regulations in Egypt: Legitimating the International Accounting Standards. *Managerial Auditing Journal* 23: 467–84. [CrossRef]

Hansen, Lars Peter. 1982. Large sample properties of generalized method of moments estimators. *Econometrica: Journal of the Econometric Society* 50: 1029–54. [CrossRef]

Hellman, Niclas, Jordi Carenys, and Soledad Moya Gutierrez. 2018. Introducing more IFRS principles of disclosure–will the poor disclosers improve? *Accounting in Europe* 15: 242–321. [CrossRef]

Houqe, Muhammad Nurul, Reza M. Monem, and Tony van Zijl. 2016. The economic consequences of IFRS adoption: Evidence from New Zealand. *Journal of International Accounting, Auditing and Taxation* 27: 40–48.

Im, Kyung So, M. Hashem Pesaran, and Yongcheol Shin. 2003. Testing for unit roots in heterogeneous panels. *Journal of Econometrics* 115: 53–74. [CrossRef]

Jacobs, Bruce I. 2009. Tumbling tower of Babel: Subprime securitisation and the credit crisis. Financial Analysts Journal 65: 17–30.

Jarque, Carlos M., and Anil K. Bera. 1987. A test for normality of observations and regression residuals. *International Statistical Review/Revue Internationale de Statistique* 55: 163–72. [CrossRef]

Jensen, Michael C., and William H. Meckling. 1976. Theory of the firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics* 3: 305–60. [CrossRef]

Judge, William, Shaomin Li, and Robert Pinsker. 2010. National adoption of international accounting standards: An institutional perspective. *Corporate Governance: An International Review* 18: 161–74.

Kao, Chihwa. 1999. Spurious regression and residual-based tests for cointegration in panel data. *Journal of Econometrics* 90: 1–44. [CrossRef]

Karavias, Yiannis, and Elias Tzavalis. 2014. Testing for unit roots in short panels allowing for a structural break. *Computational Statistics & Data Analysis* 76: 391–407.

Kasyan, Ana, Paula Gomes, Dos Santos, Carlos Pinho, and Vera Pinto. 2017. Disclosure of fair value measurement techniques of financial instruments: Study applied to the Portuguese banking sector according to IFRS 7. *International Review of Management and Business Research* 6: 1453–68.

Katz, Michael L., and Car Shapiro. 1985. Network externalities, competition, and compatibility. American Economic Review 75: 424–40.
Khan, Muhamma Yar. 2016. Corporate Governance and Cost of Capital: Evidence from Pakistani Listed Firms. Ph.D. Thesis, University of Glasgow, Glasgow, UK.

Kim, Saeron, and Haeyoun Ryu. 2018. The impact of mandatory IFRS adoption on capital markets: Evidence from Korea. *International Journal of Accounting and Information Management* 26: 38–58. [CrossRef]

- Lambert, Richar, Christia Leuz, and Rober E. Verrecchia. 2007. Accounting information, disclosure, and the cost of capital. *Journal of Accounting Research* 45: 385–420. [CrossRef]
- Leuz, Christian, and Robert E. Verrecchia. 2000. The economic consequences of increased disclosure. *Journal of Accounting Research* 38: 91–124. [CrossRef]
- Levin, Andrew, Chien-Fu Lin, and Chia-Shang James Chu. 2002. Unit root tests in panel data: Asymptotic and finite-sample properties. *Journal of Econometrics* 108: 1–24. [CrossRef]
- Li, Siqi. 2010. Does mandatory adoption of International Financial Reporting Standards in the European Union reduce the cost of equity capital? *The Accounting Review* 85: 607–36. [CrossRef]
- Lim, Chu Yeong, and See Liang Foo. 2017. Financial reporting risks in relation to financial instruments. In *The Routledge Companion to Accounting and Risk*, 1st ed. Edited by Margaret Woods and Philip Linsley. London: Routledge. Abingdon: Taylor & Francis.
- Lotfi, Chaimaa, Swetha Srinivasan, Myriam Ertz, and Imen Latrous. 2022. A tool for study on impact of Big Data technologies on firm performance. In *Intelligent Communication Technologies and Virtual Mobile Networks: Proceedings of ICICV 2022*. Lecture Notes on Data Engineering and Communications Technologies. Singapore: Springer, vol. 131.
- Lotfi, Chaimaa, Swetha Srinivasan, Myriam Ertz, and Imen Latrous. 2023. Exploring the Aggregated and Granular Impact of Big Data Analytics on a Firm's Performance through Web Scraping-Based Methodology. In SAGE Research Methods: Business & Management. Londres: SAGE Publications Ltd.
- Mager, Ferdinand, and Martin Meyer-Fackler. 2017. Mergers and acquisitions in Germany: 1981–2010. *Global Finance Journal* 34: 32–42. [CrossRef]
- Mahmoodi, Majed, and Elahe Mahmoodi. 2016. Foreign direct investment, exports and economic growth: Evidence from two panels of developing countries. *Economic Research-Ekonomska Istraživanja* 29: 938–49.
- Mantzari, Elisavet, Christos Sigalas, and Tony Hines. 2017. Adoption of the International Financial Reporting Standards by Greek non-listed companies: The role of coercive and hegemonic pressures. *Accounting Forum* 41: 185–205.
- Meyer, John, and Brian Rowan. 1977. Institutionalized organizations: Formal structure as myth and ceremony. *American Journal of Sociology* 83: 340–63. [CrossRef]
- Mohsin, Muhammad, Muhammad Nurunnabi, Jijian Zhang, Huaping Sun, Nadeem Iqbal, Robina Iram, and Qaiser Abbas. 2021. The evaluation of efficiency and value addition of IFRS endorsement towards earnings timeliness disclosure. *International Journal of Finance & Economics* 21: 1793–807.
- Nguyen, Thi Lam Anh. 2018. Diversification and bank efficiency in six ASEAN countries. *Global Finance Journal* 37: 57–78. [CrossRef] Ohlson, James A., and Beate E. Juettner-Nauroth. 2005. Expected EPS and EPS growth as determinants of value. *Review of Accounting Studies* 10: 349–65. [CrossRef]
- Opare, Solomon, Muhammad Nurul Houqe, and Tony Van Zijl. 2021. Meta-analysis of the Impact of Adoption of IFRS on Financial Reporting Comparability, Market Liquidity, and Cost of Capital. *Abacus* 57: 502–56. [CrossRef]
- Palea, Vera. 2013. IAS/IFRS and financial reporting quality: Lessons from the European experience. *China Journal of Accounting Research* 6: 247–63. [CrossRef]
- Palea, Vera. 2014. Are IFRS value-relevant for separate financial statements? Evidence from the Italian stock market. *Journal of International Accounting, Auditing and Taxation* 23: 1–17. [CrossRef]
- Pedroni, Peter. 2004. Panel cointegration: Asymptotic and finite sample properties of pooled time series tests with an application to the PPP hypothesis. *Econometric Theory* 20: 597–625.
- Persakis, Anthony, and George Emmanuel Iatridis. 2017. The joint effect of investor protection, IFRS and earnings quality on cost of capital: An international study. *Journal of International Financial Markets, Institutions and Money* 46: 1–29.
- Pesaran, M. Hashem. 2003. Estimation and Inference in Large Heterogenous Panels with Cross Section Dependence, CESifo Working Paper, No. 869. Munich: Center for Economic Studies and ifo Institute (CESifo).
- Pesaran, M. Hashem. 2006. Estimation and inference in large heterogeneous panels with a multifactor error structure. *Econometrica* 74: 967–1012. [CrossRef]
- Pesaran, M. Hashem. 2007. A simple panel unit root test in the presence of cross-section dependence. *Journal of Applied Econometrics* 22: 265–312. [CrossRef]
- Pesaran, M. Hashem. 2015. Testing weak cross-sectional dependence in large panels. Econometric Reviews 34: 1089–117. [CrossRef]
- Pesaran, M. Hashem. 2021. General diagnostic tests for cross-sectional dependence in panels. *Empirical Economics* 60: 13–50. [CrossRef] Prather-Kinsey, Jenice, Eva Jermakowicz, and Thierry Vongphanith. 2008. Capital market consequences of European firms' mandatory
- Prather-Kinsey, Jenice, Eva Jermakowicz, and Thierry Vongphanith. 2008. Capital market consequences of European firms' mandatory adoption of IFRS. Paper presented at the American Accounting Association Annual Meeting, Anaheim, CA, USA, August 3–6; pp. 1–39.
- Prather-Kinsey, Jenice, Francesco De Luca, and Ho-Tan Phat Phan. 2022. Improving the global comparability of IFRS-based financial reporting through global enforcement: A proposed organizational dynamic. *International Journal of Disclosure and Governance* 19: 330–51.
- Ramanna, Karthik, and Ewa Sletten. 2009. Why Do Countries Adopt International Financial Reporting Standards, Working Paper, Harvard Business School, Business Research for Business Leaders. Available online: www.hbs.edu/faculty/Publication%20Files/09-102\_1bc06d7-7340-4f0a-b638-e23211a40c41.pdf (accessed on 7 August 2023).

- Reichborn-Kjennerud, Kristin, Belén González-Díaz, Enrico Bracci, Thomas Carrington, James Hathaway, Kim Klarskov Jeppesen, and Ileana Steccolini. 2019. Sais work against corruption in Scandinavian, South-European and African countries: An institutional analysis. *The British Accounting Review* 51: 100842.
- Reschiwati, Reschiwati, A. Syahdina, and S. Handayani. 2020. Effect of liquidity, profitability, and size of companies on firm value. *Utopia y Praxis Latinoamericana* 6: 325–32.
- Saha, Amitav, and Sudipta Bose. 2021. Do IFRS disclosure requirements reduce the cost of capital? Evidence from Australia. *Accounting & Finance* 61: 4669–701.
- Sanjaya, I. Putu Sugiartha, May H. B. Barus, and May Hosiani. 2017. The differences cost of equity capital between before and after adoption of IFRS. *Journal Keuangan dan Perbankan* 21: 609–20.
- Sarafidis, Vasilis, and Donald Robertson. 2006. *On the Impact of Cross Section Dependence in Short Dynamic Panel Estimation*. Cambridge: University of Cambridge.
- Sharpe, William F. 1964. Capital asset prices: A theory of market equilibrium under conditions of risk. *The Journal of Finance* 19: 425–42. Smith, Adam. 1776. *An Inquiry in to the Wealth of Nations*. London: Strahan and Cadell, pp. 1–11.
- Utama, Sidharta, Aria Farahmita Mita, and Viska Anggraita. 2017. Economic consequences of IFRS adoptions around the ASEAN Countries. *International Journal of Economics and Management* 11: 529–51.
- Wook-Bin, Leem, and Jee Hoon Yuk. 2018. Effect of the IFRS adoption on the cost of capital: Evidence from Korea. *Journal of Applied Business Research (JABR)* 34: 209–16.
- Yamani, Amal, Khaled Hussainey, and Khaldoon Albitar. 2021. The impact of financial instruments disclosures on the cost of equity capital. *International Journal of Accounting & Information Management* 29: 528–51.
- Zeff, Stephen A. 2012. The evolution of the IASC into the IASB, and the challenges it faces. The Accounting Review 87: 807–37. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.